Author’s response to reviews

Title: Perceived coercion in psychiatric hospital admission: Validation of the French-language version of the MacArthur's Admission Experience Survey

Authors:
Philippe Golay (philippe.golay@chuv.ch)
Imane Semlali (imane.semlali@unil.ch)
Hélène Beuchat (helene.beuchat@unil.ch)
Valentino Pomini (valentino.pomini@unil.ch)
Benedetta Silva (benedetta.silva@chuv.ch)
Laurent Loutrel (laurent.loutrel@chuv.ch)
Jacques Thonney (jacques.thonney@chuv.ch)
Sylfa Fassassi Gallo (sylfa.fassassi@chuv.ch)
Stéphane Morandi (stephane.morandi@chuv.ch)
Charles Bonsack (charles.bonsack@chuv.ch)

Version: 2 Date: 25 Oct 2017

Author’s response to reviews:

Dr Joanne Allen
BMC Psychiatry

Dear Colleague,

We would like to submit a revised version of our manuscript “Perceived coercion in psychiatric hospital admission: Validation of the French-language version of the MacArthur’s Admission Experience Survey” for publication in BMC Psychiatry.
Please find below the detailed responses to the three issues that were raised.

We hope that this revised version will meet all the requirements for a publication in BMC Psychiatry and that this contribution will be of interest to your readers

Yours sincerely

Philippe Golay

1. Please include the email addresses of all authors on the title page of your manuscript.
   
   Answer: All email addresses have been added to the title page.

2. Please provide a List of abbreviations after the Conclusions section. If abbreviations are used in the text, they should be defined in the text at first use and included in this list.
   
   Answer: This has been done as requested.

3. Thank you for providing an Ethics and consent for participation heading, please include the full unabbreviated names of any relevant national or international regulations and guidelines followed while conducting the research presented in your manuscript.
   
   Answer: The Ethics approval and consent to participate declaration has been expanded.